

# Disentangling the impacts of anthropogenic aerosols on terrestrial carbon cycle during 1850-2014

Yuan Zhang, Philippe Ciais, Olivier Boucher, Fabienne Maignan, Ana Bastos, Daniel Goll, Thibaut Lurton, Nicolas Viovy, Nicolas Bellouin, Laurent Li

## ▶ To cite this version:

Yuan Zhang, Philippe Ciais, Olivier Boucher, Fabienne Maignan, Ana Bastos, et al.. Disentangling the impacts of anthropogenic aerosols on terrestrial carbon cycle during 1850-2014. Earth's Future, 2021, 9 (7), pp.e2021EF002035. 10.1029/2021EF002035. hal-03228346

# HAL Id: hal-03228346 https://hal.science/hal-03228346v1

Submitted on 22 Jul 2021

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



# **Earth's Future**

## **RESEARCH ARTICLE**

10.1029/2021EF002035

#### **Key Points:**

- A set of factorial simulations are set up to investigate the anthropogenic aerosol impacts on land C fluxes during 1850–2014
- Anthropogenic aerosols cumulatively enhanced land C sink by 22.6 PgC since 1850
- The large C sink increase is mainly attributed to aerosol-induced diffuse radiation changes, followed by the cooling effect

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### Correspondence to:

Y. Zhang, yuan.zhang@lsce.ipsl.fr

#### **Citation**:

Zhang, Y., Ciais, P., Boucher, O., Maignan, F., Bastos, A., Goll, D., et al. (2021). Disentangling the impacts of anthropogenic aerosols on terrestrial carbon cycle during 1850–2014. *Earth's Future*, *9*, e2021EF002035. https://doi. org/10.1029/2021EF002035

Received 10 FEB 2021 Accepted 7 MAY 2021

© 2021. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited and is not used for commercial purposes.

# Disentangling the Impacts of Anthropogenic Aerosols on Terrestrial Carbon Cycle During 1850–2014

Yuan Zhang<sup>1,2,3</sup>, Philippe Ciais<sup>1</sup>, Olivier Boucher<sup>3</sup>, Fabienne Maignan<sup>1</sup>, Ana Bastos<sup>4</sup>, Daniel Goll<sup>5</sup>, Thibaut Lurton<sup>3</sup>, Nicolas Viovy<sup>1</sup>, Nicolas Bellouin<sup>6</sup>, and Laurent Li<sup>2</sup>

<sup>1</sup>Laboratoire des Sciences du Climat et de l'Environnement (LSCE), IPSL, CEA/CNRS/UVSQ, Gif sur Yvette, France, <sup>2</sup>Laboratoire de Météorologie Dynamique, IPSL, Sorbonne Université/CNRS, Paris, France, <sup>3</sup>Institut Pierre-Simon Laplace, Sorbonne Université/CNRS, Paris, France, <sup>4</sup>Max Planck Institute for Biogeochemistry, Jena, Germany, <sup>5</sup>Université Paris Saclay, CEA-CNRS-UVSQ, LSCE/IPSL, Gif sur Yvette, France, <sup>6</sup>Department of Meteorology, University of Reading, Reading, UK

**Abstract** Aerosols have a dimming and cooling effect and change hydrological regimes, thus affecting carbon fluxes, which are sensitive to climate. Aerosols also scatter sunlight, which increases the fraction of diffuse radiation, increasing photosynthesis. There remains no clear conclusion whether the impact of aerosols on land carbon fluxes is larger through diffuse radiation change than through changes in other climate variables. In this study, we quantified the overall physical impacts of anthropogenic aerosols on land C fluxes and explored the contribution from each factor using a set of factorial simulations driven by climate and aerosol data from the IPSL-CM6A-LR experiments during 1850-2014. A newly developed land surface model which distinguishes diffuse and direct radiation in canopy radiation transmission, ORCHIDEE\_DF, was used. Specifically, a subgrid scheme was developed to distinguish the cloudy and clear sky conditions. We found that anthropogenic aerosol emissions since 1850 cumulatively enhanced the land C sink by 22.6 PgC. Seventy-eight percent of this C sink enhancement is contributed by aerosolinduced increase in the diffuse radiation fraction, much larger than the effect of the aerosol-induced dimming. The cooling of anthropogenic aerosols has different impacts in different latitudes but overall increases the global land C sink. The dominant role of diffuse radiation changes found in this study implies that future aerosol emissions may have a much stronger impacts on the C cycle through changing radiation quality than through changing climate alone. Earth system models need to consider the diffuse radiation fertilization effect to better evaluate the impacts of climate change mitigation scenarios.

**Plain Language Summary** The aerosols emitted by human activities can change climate and increase diffuse fraction of sunlight. All these changes can influence the carbon fixation of vegetation on land, further affect the atmospheric  $CO_2$  concentration and the climate. Currently, there is still no agreement on whether climate or diffuse light change is more important in affecting land carbon fixation. To solve this problem, we designed a set of experiments and used a newly developed computer code to investigate the impact of anthropogenic aerosols on land carbon sink from each climate factor and diffuse light. We found that since 1850, human-caused aerosol emissions increased the land carbon sink by about 2 years of present-day anthropogenic  $CO_2$  emissions. Seventy-eight percent of this large increase in carbon sink is mainly contributed by the increase in diffuse light fraction. The cooling caused by aerosols affected the land carbon sink differently in different latitudes and overall increased the global land carbon sink. The important role of diffuse light found here implies that aerosols emissions may have stronger impacts through changing radiation quality than through changing climate alone in the future, and climate computer codes need to consider diffuse light to better evaluate the impacts of climate change mitigation policies.

### 1. Introduction

Anthropogenic aerosols have received increasing attention during the last decades in climate change studies as they significantly affect the climate system through aerosol-radiation and aerosol-cloud interactions (Boucher et al., 2013). Besides these well-known (but not well-understood) physical processes in the atmosphere, which alter the land surface radiation budget directly, atmospheric aerosols also have a range of indirect impacts, for example, on the terrestrial C cycle (Mercado et al., 2009; Zhang et al., 2019), which in turn affects the climate system through climate-carbon feedbacks (Friedlingstein et al., 2006).

One of the main mechanisms by which aerosols affect the land C cycle is through changing the quantity and quality of short-wave radiation received by vegetation, with an impact on photosynthesis. On the one hand, the absorption and scattering of solar radiation due to atmospheric aerosols decrease the amount of incoming solar radiation at the top of the vegetation canopy. In situ observations have shown negative relationships between aerosol optical depth (AOD) and incoming solar radiation at the land surface (Kanniah et al., 2010). This decrease may cause a decrease in photosynthesis in ecosystems which are light-limited, such as tropical forests (Nemani et al., 2003). On the other hand, aerosols alter the angular distribution of incoming light. The scattering of radiation by aerosols significantly increases the fraction of diffuse light, which is able to penetrate deeper and distribute more evenly in the canopy (Roderick et al., 2001). Since sunlit leaves are often light-saturated while shaded leaves are not, an increase in diffuse radiation may increase the photosynthesis of the entire canopy. In situ observations at ecosystem level show that diffuse radiation increases the light use efficiency (LUE) of forests, grasslands, and croplands (Alton, 2008; Alton et al., 2007; Choudhury, 2001; Gu et al., 2002; Healey et al., 1998).

The compensation issue of gross primary production (GPP, the photosynthesis rate) gain due to diffuse radiation increase procured by aerosols and clouds and GPP loss due to total radiation decline has been investigated at the site level (Alton et al., 2007; Hollinger et al., 1994; Kanniah et al., 2010) in the past. There is, however, no consensus on the quantitative role of each process. One reason is that the field measurements were performed in various ecosystems under different environmental conditions, highlighting the importance of understanding this question at a larger scale. Another reason is that the precise effects of decreased light and increased diffuse light fraction depend on the amount of aerosols in the atmospheric column, their optical properties, and the relative occurrences of clear and cloudy sky conditions.

At larger scales, the aerosol impacts are generally investigated using atmospheric and land surface models (LSM) (Jones et al., 2003; Mahowald et al., 2011; Zhang et al., 2019). These studies rely on simulations with and without anthropogenic aerosols and estimate the aerosol impacts as the difference between such pairs of simulations. However, due to the lack of representation of direct and diffuse radiation in canopy light transmission in LSMs, the aerosol impacts investigated by the above studies did not include the fertilization effect of diffuse radiation. Mercado et al. (2009) addressed this problem by modifying the canopy light transmission scheme in the Joint UK Land Environment Simulator (JULES) model to represent the differential absorption of diffuse and direct radiation. They quantified the aerosol impacts on the land C sink using this new scheme driven by observation-based climate fields. Their results strictly presented the impacts of aerosols through changing radiation, but did not consider the impacts of aerosol-caused cooling and precipitation changes, which were suggested to also have considerable impacts (Jones et al., 2003; Mahowald et al., 2011; Zhang et al., 2019). Recently, Xie et al. (2020) compared the impacts of aerosol-induced diffuse radiation fraction, temperature, and vapor pressure deficit changes on vegetation productions in China, and found a dominant impact from the changes in diffuse light fraction. However, their study omitted the impacts of aerosol-induced dimming and precipitation changes. Also, their study area is restricted to an industrialized region with strong aerosol emissions, which is not very representative of the entire land area. Therefore, the need still remains to investigate and quantify the full impact of anthropogenic aerosols on the land C budget at the global scale.

Currently, there is no standard way to deal with diffuse radiation and to assess its impacts. Existing LSMs either use empirical equations to partition total radiation into its direct and diffuse components (Zhang et al., 2020), or consider the weighted average of radiation in the clear and cloudy fractions of a gridbox (Mercado et al., 2009). In both approaches, gridboxes are considered to be homogeneous and to receive single values of total radiation and diffuse radiation fraction. However, the averaging of diffuse radiation fraction over a LSM gridbox (~55 km or larger) and time step (half hourly or longer) can potentially bias land C fluxes (Zhang et al., 2021). Furthermore, LSMs with a representation of diffuse light were evaluated using site level observations (Mercado et al., 2009; Zhang et al., 2020), where cloudy and clear conditions do not coexist at the same time step. Therefore, simulating C fluxes in a heterogeneous gridbox is a complex task. We argue that it is necessary to at least distinguish clear-sky and cloudy conditions within a model gridbox.



For the above reasons, we present here a new study to quantify the impacts of anthropogenic aerosols on land C fluxes during the historical period (1850–2014) using a LSM that simulates photosynthesis in the clear and cloudy gridbox fractions separately. A set of factorial simulations based on the fields of the IPSL-CM6A-LR climate model is designed to investigate the total impact of anthropogenic aerosols and to separate the impacts of each aerosol-mediated mechanism including changing (a) radiation quality, (b) radiation quantity, (c) temperature, and (d) precipitation. The study period is the historical period from 1850 to 2014, during which anthropogenic emission aerosols have increased strongly and exerted a significant effect on climate change. The objectives of this study are to (a) quantify the impacts of aerosol-induced changes of diffuse radiation and climate change on the land C budget during the historical period; (b) determine whether the fertilization by diffuse radiation globally compensates for the effect of reduction of incoming radiation due to anthropogenic aerosols; (c) understand how aerosol-induced changes in radiation, temperature, and precipitation individually affect land C fluxes and the historical carbon budget.

#### 2. Data and Methods

#### 2.1. ORCHIDEE\_DF Subgrid Version

In order to investigate the full impacts of aerosols including the diffuse radiation fertilization effect on the land C fluxes, we performed simulations with ORCHIDEE\_DF, a new development of ORCHIDEE trunk (v5453) (Krinner et al., 2005; Zhang et al., 2020). ORCHIDEE\_DF has a two-stream canopy light transmission model which distinguishes direct and diffuse radiation and considers the differential light absorption of sunlit and shaded leaves. ORCHIDEE\_DF has been intensively evaluated over 159 flux sites and proved to be able to capture the observed diffuse radiation fertilization effect (Zhang et al., 2020). A detailed description of ORCHIDEE\_DF can be found in Zhang et al. (2020).

As stated in Section 1, the homogenization treatment of clear and cloudy fractions in a LSM gridbox may introduce biases in the C flux simulation. To avoid the issue, we modified ORCHIDEE\_DF to calculate photosynthesis of each plant functional type twice in each gridbox, for example, under clear sky and cloudy sky conditions, respectively. The clear sky and cloudy sky photosynthesis rates are then weighted by cloud cover and integrated to grid level GPP (Figure 1). To realize this subgrid simulation, the incoming shortwave radiation and diffuse radiation fraction of each clear and cloudy subgrid are needed. These subgrid variables are usually not available in climate models, while the total incoming shortwave radiation, cloud fraction, and aerosol optical depth per gridbox can be often found, especially in data sets from Earth system models (ESMs). Using these grid level variables, we developed a module to obtain the missing subgrid clear and cloudy sky radiation and diffuse radiation fraction in ORCHIDEE\_DF (Figure 1). How we partition radiation and calculate diffuse radiation fraction is detailed in Appendix A.

#### 2.2. Forcing Data and Experimental Design

The climate forcing in this study is derived from two simulations performed with the IPSL-CM6A-LR (Boucher et al., 2020), an ESM consisting of an atmospheric model LMDZ (v3554), an ocean model NEMO (v9455), a land surface model ORCHIDEE (trunk v5661), and a coupler (OASIS). As a model participating to the Coupled Model Intercomparison Project Phase 6 (CMIP6, Eyring et al., 2016), IPSL-CM6A-LR ran a preindustrial control simulation (piControl) driven by constant external forcing representative of the year 1850, after a long spin up. The historical climate fields are taken from the IPSL-CM6A-LR historical r1i1p1f1 simulation, which is one of the simulations performed for the 1850–2014 period driven by all natural and anthropogenic forcing and initialized from the January 1, 1910 state of the piControl simulation. From the same initial conditions of the piControl simulation, another simulation, hist-piAer (r1i1p1f1, hereafter referred to as histNoAA) was performed by prescribing all the forcing terms to be the same as in historical r1i1p1f1 but keeping the anthropogenic aerosols at pre-industrial level (1850).

The two simulations provide daily climate surface or near-surface variables including 2-m maximum (*T*max) and minimum (*T*min) air temperature, precipitation (Precip), incoming shortwave radiation at land surface (SWdown), incoming longwave radiation at land surface (LWdown), near surface specific humidity (Qair), near surface air pressure (*P*surf), 10-m eastward wind component (Wind\_E), and 10-m northward wind component (Wind\_N). As the modeled climate may be biased from the observations, causing errors





**Figure 1.** The sub-grid GPP calculation in this study. clt, cloud fraction; Fdf, fraction of diffuse shortwave radiation; SWdown, downward shortwave radiation at land surface; od550aer, tropospheric AOD at 550 nm; od550\_STRAT, stratospheric AOD at 550 nm. AOD, aerosol optical depth.

in C fluxes in simulations, we introduced a bias correction procedure using the observation-based CRU-JRA v1.1 data set as the reference to ensure the climate used in this study has the same baseline as the observation-based data set for the preindustrial period (1850–1859). This method has been used in Zhang et al. (2019) and is described in Appendix B. The aerosol information was obtained from the monthly tropospheric aerosol optical depth at 550 nm (od550aer) and monthly stratospheric aerosol optical depth at 550 nm (od550\_STRAT) as used to drive the IPSL-CM6A-LR ESM simulations. Because clouds change on short (subdiurnal) timescales and have a strong impact on diffuse radiation, 3-h total cloud fraction (clt) data from the same IPSL-CM6A-LR simulations is used. Due to the lack of observations of preindustrial aerosol and cloud properties and amounts, we did not perform the same bias-correction on aerosol and cloud fraction variables. All these variables which were originally provided on a 144 × 143 global grid were first re-gridded into a  $2^{\circ} \times 2^{\circ}$  grid as model input, and further interpolated to 30-min time steps in ORCHIDEE\_DF. The climate variables are interpolated using the default method in ORCHIDEE LSM to generate reasonable diurnal cycle, and the aerosol optical depth and cloud fraction are interpolated linearly.

For the LSM simulations, to get equilibrated initial C pools, a spin-up simulation using the 1850–1869 bias-corrected climate from the historical scenario ESM simulation (hereafter all climate refers to the bias-corrected climate if not specified) was first performed for 400 years using the spin-up analytic mode of ORCHIDEE (Lardy et al., 2011), which accelerates the reach of equilibrium state for slow soil carbon pools. After the spin-up, a set of LSM experiments separating the impacts of anthropogenic aerosols were set up. The detailed information about the experimental design can be found in Table 1.



#### Table 1

Experimental Design for the IPSL Simulations

|            | Forcing source    |        |        |   |     |  |   |   |                    |
|------------|-------------------|--------|--------|---|-----|--|---|---|--------------------|
| Name       | Tair              | Precip | SWdown | Tropospheric<br>AOD<br>(partition SW<br>between clear<br>and cloudy<br>sub-grid area) | clt | Tropospheric AOD (us<br>SWdown in clea | sed to calculate diffuse<br>ar sub-grid area) | Stratospheric AOD (almost same for hist and NoAA) | Other<br>variables |
| Hist       | Hist <sup>a</sup> | Hist   | Hist   | Hist  |     | hist                                   | Hist  | hist  | hist               |
| NoAA       | NoAA <sup>a</sup> | NoAA   | NoAA   | NoAA  |     | NoAA                                   | NoAA  | hist  | NoAA               |
| NoAASW     | Hist              | Hist   | NoAA   | NoAA  |     | NoAA                                   | NoAA  | hist  | hist               |
| NoAAprecip | Hist              | NoAA   | hist   | hist  |     | hist                                   | hist  | hist  | hist               |
| NoAATair   | NoAA              | Hist   | hist   | hist  |     | hist                                   | hist  | hist  | hist               |
| NoAARqlt   | Hist              | Hist   | hist   | hist  |     | hist                                   | NoAA  | hist  | hist               |
| NoVA       | Hist              | Hist   | hist   | hist  |     | hist                                   | Hist  | 0 all the time <sup>a</sup>                       | hist               |

<sup>a</sup>The stratospheric AOD values under most volcano-free cases are grouped into the same category as 0 in the look up table.

1,2 hist and NoAA in the forcing field of the table indicate historical (r1i1p1f1) and hist-piAer (r1i1p1f1) from IPSL-CM6A-LR.

The hist and the NoAA LSM simulations were respectively driven by the full set of climate variables from the two ESM simulations described above. The difference between the two simulations corresponds to the full impact of anthropogenic aerosols due to both their effects on climate and diffuse light fraction. As radiation, precipitation, and temperature are the three most important factors controlling land carbon fluxes (Nemani et al., 2003), we performed the NoAASW, the NoAAprecip, and the NoAATair LSM simulations, which use the historical ESM climate but replace the SWdown, the precipitation and the Tair fields, respectively, by their counterpart from the histNoAA scenario. These factorial simulations aim to investigate the relative contributions of the aerosol effects from each one of these climate variables. The difference between the hist and the NoAASW simulations represents the impacts of the aerosol-induced full change in solar radiation, including both quality and quantity. Beside these experiments, we also set up the NoAARqlt LSM simulation, which used the same forcing as the hist simulation but recalculated the diffuse light fraction of the clear area of each gridbox using the tropospheric AOD from the histNoAA simulation. This simulation is used to investigate the impacts of the fraction of diffuse radiation (Fdf) change caused by anthropogenic aerosols alone. In the end, a LSM simulation without stratospheric aerosols from volcanoes (NoVA) was performed to investigate the volcanic impacts alone. It should be noted that the difference between the hist and the NoVA simulations includes only the impacts of the Fdf change from volcanoes but not the full climate impact caused by volcanic aerosols because the other climate fields driving the two simulations are the same. All the LSM simulations used the annual land cover and land-use fields from the ESA-LUH2v2 data set.

### 3. Results

#### 3.1. Aerosol-Induced Climate Change

The variation of bias-corrected global land yearly mean climate from the IPSL-CM6A-LR ESM *historical* and *histNoAA* experiments is shown in Figure 2. As the emission of anthropogenic greenhouse gases and aerosols did not significantly increase before the early part of the 20th century, the climate remains relatively stable during this period and the two scenarios show no substantial difference. For both scenarios, the land mean temperature, precipitation, and SWdown generally vary around global averages of  $13^{\circ}$ C, 800 mm yr<sup>-1</sup>, and 196 W m<sup>-2</sup>, respectively. The average global cloud cover over land regions is around 51%–52%, and the land-averaged tropospheric AOD average stays around 0.1 before 1940. The global stratospheric AOD is smaller than 0.01 in volcano-free years and can increase by 10-fold during large volcanic eruptions, for example, Krakatoa in 1883, El Chichón in 1982, Mount Pinatubo in 1991 (Figure 2f). After 1950, as a result of the increased anthropogenic aerosol emissions over industrialized areas (Figure 2e), the climates under





**Figure 2.** The bias-corrected annual-mean climate temporal evolution over terrestrial regions from the historical (blue lines) and histNoAA (red lines) simulations (a)–(e). The red line in (f) represent a scenario with no volcanic eruptions. (a) Air temperature; (b) Precipitation; (c) Downward shortwave radiation; (d) Total cloud fraction; (e) Tropospheric aerosol optical depth; (f) Stratospheric AOD. AOD, aerosol optical depth.

the two ESM scenarios start to diverge. For Tair (Figure 2a), both the historical and histNoAA Tair show an increase after the 1950s. However, the Tair under the histNoAA scenario increases faster, resulting in a  $0.6^{\circ}$ C warmer climate during the last 2 decades compared with the historical scenario. In terms of SWdown, a dimming effect is found along with the cooling from anthropogenic aerosols (Figure 2c). The historical SWdown decreases by about 3 W m<sup>-2</sup> since the 1950s, while the histNoAA SWdown shows no significant decrease during the same period. Compared with Tair and SWdown, no substantial differences are found on precipitation and cloud fraction between the historical and the histNoAA scenarios.

The spatial distributions of the ESM-simulated aerosol-induced climate changes are shown in Figure 3. Considering that the historical and the histNoAA scenarios differ significantly only after 1950s regarding tropospheric AOD (Figure 2e), we focus on the period 1961–2010 to investigate the aerosol impact. The Tair from IPSL-CM6A-LR indicates a widespread cooling effect of anthropogenic aerosols over land regions (Figure 3a) except western North America, where the mean *T*air values are not significantly different between the two scenarios. In contrast to the ubiquitously lower *T*air, changes in precipitation and an aridity index (precipitation-potential evapotranspiration [P-PET]) are more heterogeneous (Figures 3b and 3f). A slight to medium drying effect of aerosols is found in central Africa, India, South China, and eastern US, while in Argentina and regions around the Mediterranean, anthropogenic aerosols are found to have a wetting effect. Compared with the precursor version of the ESM (IPSL-CM5A-LR) climate (Zhang et al., 2019), the climate from IPSL-CM6A-LR does not show a strong drying effect due to aerosols in Amazonia, while it neither shows a significant wetting effect of aerosols in this region, as detected by the other two CMIP5 models used in Zhang et al. (2019).

In terms of SWdown changes (Figure 3c), a dimming effect of anthropogenic aerosols is found in 78% continental regions. The magnitude of the dimming is generally 0–5 W m<sup>-2</sup> except over industrialized regions: East Asia, Europe, and East US, where the dimming is over 5 W m<sup>-2</sup>. A 0–3 W m<sup>-2</sup> brightening is found in





**Figure 3.** The spatial distribution of anthropogenic aerosol-induced climate changes (historial-histNoAA) during 1961–2010. (a) Changes in Tair; (b) changes in precipitation; (c) changes in SWdown; (d) changes in cloud fraction; (e) changes in tropospheric AOD; (f) changes in an aridity index calculated as the difference between precipitation and PET. AOD, aerosol optical depth; PET, potential evapotranspiration.

Alaska, East Russian, Central Africa, and Australia, where aerosols in one region have resulted in dynamical changes leading to fewer clouds and more incoming solar radiation (Figure 3d).

As expected, emissions of anthropogenic aerosols and their precursors cause an increase of tropospheric aerosol optical depth, with the largest increase found in the most polluted regions (East Asia, India, Europe, and eastern US) (Figure 3e). It should be noted that the emissions of anthropogenic aerosols do not always increase tropospheric AOD. For instance, a lower tropospheric AOD is found in Central Africa under the historical scenario compared with the histNoAA scenario (Figure 3e). This might be due to an altered aerosol distribution in response to aerosol-induced atmospheric circulation changes.

#### 3.2. Response of Land C Fluxes to Aerosols

The evolution of GPP and total ecosystem respiration (TER, the sum of autotrophic and heterotrophic respiration) from different LSM experiments are shown in Figure 4. All the experiments show a GPP increase from 124–125 PgC yr<sup>-1</sup> in 1850 to about 165–168 PgC yr<sup>-1</sup> in 2014 (Figure 4a). This GPP increase is probably, to a large extent, due to the fertilization effect of rising atmospheric CO<sub>2</sub> concentration (Piao et al., 2013). Despite this consistent trend, differences among simulations remain significant. Compared with the NoAA simulation, the hist simulation has lower GPP in most of the years before the 1960s, but higher GPP afterward (Figure 4c), indicating that the response of GPP to changing anthropogenic aerosols has changed by the 1960s, when aerosol emissions started to grow quickly (Figure 2). Before 1960, the GPP difference between the hist and the NoAA simulations is very similar to the impacts of Tair alone (hist-NoAATair)





**Figure 4.** The evolution of total land (a) GPP and (b) TER in each simulation and the difference between the hist and different factorial simulations (c) and (d) to the impact of aerosols. The differences (c) and (d) have been smoothed using a running average of 5-year window to decrease the noise. GPP, gross primary production; TER, total ecosystem respiration.

(Figure 4c), while changes in SWdown or precipitation alone (NoAASW and NoAAprecip compared with hist) have small impacts on global GPP. This result indicates that the aerosol impacts on GPP is dominated by temperature changes before 1950s. However, after the 1960s, the full aerosol impact (hist-NoAA) on GPP starts to deviate from the temperature-alone induced change (hist-NoAATair) and becomes positive. This change is related to the increasing impact of aerosol-induced SWdown changes, as the impact from aerosol-induced precipitation changes remains small but the GPP difference between the hist and the NoAASW simulations increases strongly from <1 to 3 PgC yr<sup>-1</sup> during this period. In addition, the NoAARqlt simulation shows very similar GPP compared to the NoAASW simulation, implying that the GPP change is more attributable to the change in radiation quality (diffuse radiation fraction) rather than quantity (dimming). By comparing GPP from the NoVA and the hist simulations, we are able to better understand the impact of the Fdf change caused by volcanic aerosols alone. For the years with strong volcanic eruptions, GPP increases of 5–10 PgC yr<sup>-1</sup> can be found in all simulations except in NoVA (Figure 4a). This difference is generally larger than the slowly varying and weaker impacts of anthropogenic aerosols, however, it lasts for a short time. The stratospheric AOD can increase by over 0.15 after large eruptions, such as Pinatubo in 1991 (Figure 2f). This is larger than the tropospheric AOD increase caused by anthropogenic aerosols (Figure 2e). For years without volcanic eruptions, the hist and the NoVA GPP are generally identical, as expected.

In terms of TER, the global averages show similar increasing trend among the factorial simulations (Figure 4b). From 1850 to 2014, the global TER increases from 119–120 to 150–153 PgC yr<sup>-1</sup>. This increase should be mainly due to the increase in GPP, which provides more organic matter for respiration, and the increase in temperature, which accelerates the turnover of soil C pools. The difference in TER between the hist and the other experiments shows very similar patterns as the difference in GPP (Figures 4c and 4d). Before the 1950s, anthropogenic aerosols cause a 0–1 PgC yr<sup>-1</sup> decrease in TER mainly through their impact on temperature (cooling, causing a reduced GPP). While after the 1960s, a positive impact of aerosol-caused SWdown changes (quality and quantity) on TER is found and the total TER response to aerosols finally becomes positive after the 2000s. The impacts of volcanic aerosols on TER are smaller but last much longer than GPP. After each eruption, following a sudden increase, a gradual decline of TER can be found during the following 3–4 decades or until the next volcano eruption (Figure 4d).





Figure 5. The cumulative land NBP difference between hist and different factorial simulations. (a) Historical evolution of NBP difference; (b) Cumulative NBP since 1850 at the end (2014) of each simulation, numbers in color indicate the cumulative NBP difference caused by aerosols through changing each climate variable. NBP, net biome production.

The cumulative net biome production (NBP, the net C flux, calculated as GPP–TER–C release from natural and anthropogenic disturbances) changes caused by aerosols are displayed in Figure 5. As a result of the decreased GPP due to aerosols before 1950s, aerosol-induced NBP change (hist-NoAA) is negative (anthropogenic aerosols decrease land C sink) before the 1950s. After 1950s, the cumulative NBP difference becomes positive and grows quickly. At the end of the study period (2014), the hist simulation shows a 22.6 PgC higher cumulative NBP than the NoAA simulation (Figure 5b), indicating a large positive impact of anthropogenic aerosols on the cumulative land C storage during the historical period.

Besides the full impact of anthropogenic aerosols, we also investigated the impact of each environmental factor on cumulative NBP in Figure 5 by comparing the hist simulation and each factorial simulation. In comparison to hist, the cumulative NBP from the NoAATair simulation decreases before 1880s and stays around -6 PgC until 1950s (Figure 5a). An investigation on the historical and histNoAA climate shows that during the first two decades, the historical climate is much cooler than histNoAA in Eurasia, leading to smaller GPP in the hist simulation (Figure S1). Given that the radiative forcing from anthropogenic aerosol emissions remains small at global scale during this period, the large difference in temperature between ESM simulations historical and histNoAA should be interpreted as ESM's internal variability (most likely stochastic) triggered by the change in implementation of aerosol emissions in the two ESM simulations. After the 1950s, the impact of aerosol-induced Tair changes reversed and the difference in cumulative NBP increased to a positive value of 3.6 PgC in 2014.

Compared with the impacts of Tair, the impacts of precipitation on cumulative NBP (hist-NoAAprecip) show much smaller impacts, within 0–3 PgC during the entire study period (Figure 5a), indicating limited impacts of aerosol-induced precipitation changes.

In terms of radiation, the cumulative NBP differences between the hist and the NoAARqlt is small before the 1950s, indicating limited impacts of aerosol-induced radiation quality changes during this period. This small impact is not surprising as the anthropogenic aerosol-induced change in light quality is small during this period (Figures 2d–2f). After 1950s, along with the increasing aerosol emissions, strong increases in cumulative NBP difference between the hist and the NoAARqlt simulations are found (Figure 5a). The cumulative NBP difference between the hist and the NoAASW is largely consistent with the difference between the hist and the NoAASW is largely consistent with the difference between the hist and the NoAASW is largely consistent with the difference between the hist and the NoAASW is largely consistent with the difference between the hist and the NoAASW is largely consistent with the difference between the hist and the NoAASW) and diffuse light fraction changes alone (hist-NoAARqlt) enhanced the cumulative global land C sink by 14.7 and 17.7 PgC, respectively. As the total radiation changes include changes in light quantity (total SWdown), cloud fraction and the diffuse light fraction, the aerosol-induced changes in light quantity and cloud fraction have inhibited the cumulative land C sink by 3.0 PgC. Considering the full impacts of anthropogenic aerosols of 22.6 PgC, the changes in diffuse light fraction alone contributed to about 78% of the increase in land C sink from anthropogenic aerosols. If we only exclude the first 30 years when the climate internal variability shows a large impact on C fluxes and consider the period





**Figure 6.** The cumulative land NBP difference between the hist and different factorial simulations in different latitudes. (a) 60°–90°N; (b) 30°–60°N; (c) 30°S–30°N, and (d) 30°–60°S. The vertical lines denote the years with main volcanic eruptions during the study period, from the left to right: Krakatoa, Santa María, Mount Agung, El Chichón, and Mount Pinatubo. NBP, net biome production.

after 1880 (Figure S4), the impact from diffuse light fraction (17.6 PgC) still explains 66% of the full aerosol impact (26.5 PgC).

Unlike anthropogenic aerosols, volcanic aerosols are emitted sporadically during years with large eruptions. In response, the immediate (1–2 years after eruption) land C sink can surge by over 6 PgC depending on the intensity of the eruption (Figure 5a). These anomalies do not disappear right after the periods during which the stratosphere is influenced by the volcanic eruptions. Instead, the cumulative NBP difference between the hist and the NoVA simulation retains a memory and relaxes slowly toward zero during the following decades, at least until the next eruption takes place. In 2014, the diffuse radiation changes caused by volcanic aerosols cumulatively increased the cumulative land C sink by 3.9 PgC, which is less than the persistent changes in diffuse light from anthropogenic aerosols.

#### 3.3. Regional C Flux Changes

To further understand how different factors affected the aerosol impacts regionally, we further compared the NBP difference between the hist and the factorial simulations in different latitudes (Figure 6).

In the northern high latitudes (>60°N), all factorial simulations except the NoAATair show small cumulative NBP difference (<0.6 PgC) compared with the hist simulation (Figure 6a), indicating small impacts of aerosol-induced radiation and precipitation changes in this region. In contrast, the temperature-alone changes from anthropogenic aerosols cause a remarkable decline in cumulative NBP (hist-NoAATair) over those temperature-limited ecosystems during the entire study period. This decline is small before the 1950s but becomes much more pronounced afterward, resulting in a 2.3 PgC lower cumulative C sink in the hist than in the NoAATair simulation by 2014. The full impact of anthropogenic aerosols (hist-NoAA) shows very similar trend and interannual variation compared with the impact of aerosol induced temperature changes only (hist-NoAATair), implying that anthropogenic aerosols are mainly causing a decline in C sink through their cooling effect in this region.

In contrast to the northern high latitudes, the northern mid latitudes (30°–60°N) show a positive impact of anthropogenic aerosols (hist-NoAA) on cumulative NBP (Figure 6b). The positive impact of aerosols

(hist-NoAA) is generally in line with impacts exerted by SWdown (hist-NoAASW) or by only diffuse radiation fraction (hist-NoAARqlt), slightly offset by the negative impact of aerosol-induced Tair changes (hist-NoAATair), indicating that the impact of aerosols in this region is mainly through the changes in light quality offsetting the effect of cooling. The difference between the hist and the NoAAprecip simulations remains small in this region. Until the end of the study period in 2014, anthropogenic aerosols cumulatively increased NBP in this region by 8.5 PgC. SWdown, Fdf, and Tair, respectively increased NBP by 9.3, 8.9 PgC and decreased NBP by 2.4 PgC.

Unlike the northern mid and high latitudes, where aerosol induced NBP changes are dominated by a single environmental factor, no single factor is found to explain the full impacts of anthropogenic aerosols at low latitudes (30°S–30°N, Figure 6c). In this region, the cumulative NBP differences between the hist and each of the NoAA, the NoAATair, and the NoAASW simulations decrease in the first half of the study period and increase rapidly after the 1950s. Until 2014, the overall effect of anthropogenic aerosols increases the cumulative C sink in this region by 14.9 PgC, with changes in Tair, Fdf, and SWdown explaining 8.5, 8.1, and 3.9 PgC, respectively. Note that in the tropics, cooling induced by aerosols has a positive effect on NBP as it reduced the evaporative demand and allows more photosynthesis, unlike in the mid and high latitudes where biomes are mainly temperature limited. Compared with other variables, the impacts of aerosol-induced precipitation changes are small for the low latitudes.

The impacts of aerosols on the land C sink in the southern mid latitudes  $(30^{\circ}S-60^{\circ}S)$  is smaller than in other latitudes because the land area in this region is relatively small (Figure 6d). Until the end of the study period, the anthropogenic aerosols enhanced the C sink by 1.0 PgC (hist-NoAA), which is close to the single impact of aerosol-induced precipitation changes (hist- NoAAprecip, 1.0 PgC) and slightly smaller than the impact of SWdown changes (hist-NoAASW, 1.5 PgC). The changes in Tair cause a small decline of cumulative NBP (0.5 PgC) before 1900 and an increase of C sink after 1960s. At the end of the study period, the difference between hist and NoAATair is only -0.3 PgC in cumulative NBP. It should be noted that unlike the other latitudes where the impact of radiation quality changes is larger than the full impacts of radiation changes, in this region, the cumulative NBP difference between hist and NoAASW. This is because the anthropogenic aerosol emissions are small in this region and have little impact on tropospheric AOD and thus diffuse radiation fraction (Figure 3e). In addition, the SWdown increased in a considerable proportion of the land area in this region in response to anthropogenic aerosol-induced cloud fraction changes (Figures 3c and 3d), this brightening may be responsible to the large difference in radiation impacts between this region and the other latitudes.

#### 3.4. Spatial Distribution of the Aerosol Impacts

Since the aerosol-induced flux changes mainly occur after 1950s, to further understand how this happens, we examine the detailed regional spatial patterns of GPP and NBP differences between the hist and each factorial simulation during 1961–2010 in Figures 7 and 8.

The full impacts of anthropogenic aerosols on GPP generally show a latitudinal pattern (Figure 7a), with most regions in low latitudes experiencing a positive impact of anthropogenic aerosols and most mid to high latitude regions a negative impact, except for East Asia, eastern US, and Europe, where the most pronounced tropospheric AOD increases in response to anthropogenic aerosol emissions are found (Figure 3). The strongest GPP decrease in response to aerosols is found in central and western US and northern Russia, with a magnitude of over 50 gC m<sup>-2</sup> yr<sup>-1</sup>.

The impacts of aerosol-induced Tair changes show a similar latitudinal pattern as the full aerosol impacts (Figure 7b), that is, a decreased GPP at mid to high latitudes and an increased GPP at low latitudes. In contrast to the full impacts of anthropogenic aerosols, the Tair changes alone from aerosols generally decreased GPP in East Asia, eastern US, and Europe. In contrast to Tair, aerosol-caused precipitation changes have relatively small impacts on GPP in most regions, except for semi-arid regions (Figure 7c). In over 85% of the land area, the precipitation changes cause less than 10 gC m<sup>-2</sup> yr<sup>-1</sup> changes in GPP (generally <1% of annual GPP). Nevertheless, in relatively dry regions like the Sahel and Central India, GPP decreases significantly in response to aerosol-induced precipitation changes (Figure 7c), which partly explains the total aerosol-induced GPP decline in these regions (Figure 7a). Similar positive responses of GPP to aerosol-induced

## Earth's Future





**Figure 7.** The spatial distribution of GPP difference between hist and each factorial simulation during 1961–2010 (a–e), and between NoAARqlt and NoAASW (f, representing the effect of radiation quantity and cloud cover). GPP, gross primary production.

precipitation changes are found in the semi-arid regions in southern South America and southern Africa. The impact of radiation (Figures 7d and 7e) changes translates into a spatial pattern strongly related to the anthropogenic aerosol emissions (Figure 3). Both NoAASW and NoAARqlt have over 50 gC m<sup>-2</sup> yr<sup>-1</sup> higher GPP than hist in the most polluted regions, that is, East Asia, eastern US, and Europe. Meanwhile, the dimming due to aerosols tends to decrease the GPP in these polluted regions (Figure 7f). Generally, the impact of aerosol-caused overall radiation changes on GPP are more dominated by changes in radiation quality (Figures 7d and 7e), except regions in India, Southeast Asia, and Amazon, where GPP is decreased due to the changes in light quantity (mainly dimming) and cloud fraction (Figures 7d and 7f). The simulated light limitation of GPP at low latitudes is in line with analysis of climatic data and satellite observations of vegetation activity by Nemani et al. (2003).

The NBP response to anthropogenic aerosols generally shows a similar spatial pattern as that of GPP, with a positive response in low latitudes and a negative response in high latitudes (Figure 8a). Compared to GPP, the area of NBP decrease in mid and high latitudes is much smaller. Large regions in North America and Russia show negative impacts of anthropogenic aerosols on GPP but positive impacts on NBP (Figures 7a and 8a). These patterns are in line with the Tair change-induced C flux changes (Figures 7b and 8b), that is, at the mid latitudes, aerosol-induced cooling enhances the regional land C sink by decreasing more respiration than photosynthesis, which is consistent with a previous study using simulations driven by CMIP5 climate data (Zhang et al., 2019). In contrast to temperature, which has strong impacts on heterotrophic respiration, the aerosol-induced radiation changes do not directly alter respiration, but only indirectly through litter input. Not surprisingly, the spatial pattern of radiation change-induced GPP and NBP changes are

## Earth's Future







similar (Figures 7d–7f and 8d–f). NBP is found significantly increased in response to the radiation changes in forested regions with high aerosol emissions such as East and South Asia, eastern US, and Europe. The responses of NBP to aerosol-induced precipitation changes are small on the 50-years time scale (Figure 8c).

#### 3.5. Aerosol Impacts on GPP for Clear and Cloudy Days

Because under cloudy condition the Fdf is always 1, the impacts of aerosols on C fluxes strongly depends on whether the sky is clear or cloudy. Figure 9 clearly illustrates this difference. In clear days (Figure 9a),







anthropogenic aerosols are found to enhance the GPP of most land regions, especially of the most industrialized regions including Europe, India, East Asia, and East US. The maximum GPP difference can be over 200 gC m<sup>-2</sup> yr<sup>-1</sup>, which is found in East Europe. Some exceptional negative impact on GPP is found in high latitude where ecosystems are limited by temperature. In contrast to the clear days, the impact of aerosols on GPP in cloudy days show a meridional pattern with negative impacts in mid and high latitudes and positive impacts in tropical regions (Figure 9b), which is generally consistent with the Tair-alone caused GPP changes (Figure 7b). Some exceptions are found in India with negative GPP impact of over -150 gC m<sup>-2</sup> yr<sup>-1</sup>. This negative impact might be attributed to strong decline of SWdown during cloudy days in this region (Figure S2), possibly due to aerosol-induced circulation changes. Because NBP has a lagged response to GPP, the aerosol-induced NBP changes under cloudy and clear days are much noisier than the GPP changes (Figure S3).

## 4. Discussion

#### 4.1. Global Impacts of Aerosols on C Fluxes

Anthropogenic aerosols are found to decrease the simulated global GPP by 0.22 PgC yr<sup>-1</sup> during the entire study period (1850–2014). Along with the decline in GPP, anthropogenic aerosols also induced a decrease of global TER of 0.37 PgC yr<sup>-1</sup>, which is stronger than the GPP decrease, leading to a global increase of land C sink by 0.14 PgC yr<sup>-1</sup>, or cumulatively 22.6 PgC since 1850. This increase of land C sink is about one half of the cumulative NBP (51 PgC) during this period and can compensate about 2 years of anthropogenic  $CO_2$  emissions during the last decade (Friedlingstein et al., 2014). It should be noted that there is a cumulative NBP difference of about 6 PgC before 1880 due to a particular cold period of 2–3 decades at the beginning of the ESM hist simulation, attributable to ESM's internal variability (Figures 5 and S1). If we account for the anthropogenic aerosol impacts since 1880 and masked out the period before, the cumulative aerosol impacts on NBP would be larger, with the magnitude of 26.5 PgC (Figure S4).

When considering only the period after the 1950s, we found an increase of GPP in response to anthropogenic aerosols 0.27 PgC yr<sup>-1</sup>, which differs significantly from the GPP decline found before the 1950s. This reversed pattern is very different from a recent study using a comparable approach but omitting the changes in light quality omitted (Zhang et al., 2019). They showed that aerosol impacts on GPP did not change much before and after the 1950s. In our model, the changes in diffuse radiation after the 1950s have significantly altered the aerosol impacts.

Among the large overall impact of anthropogenic aerosols, the changes in diffuse radiation fraction accounted for a 17.7 PgC (17.6 PgC after 1880) larger cumulative land storage. This large enhancement of land C storage due to light quality change alone indicates that the diffuse radiation fertilization effect reported at flux sites for short term periods (Alton et al., 2007; Cheng et al., 2015; Gu et al., 2002; Niyogi et al., 2004; Misson et al., 2005; Kanniah et al., 2013; Knohl & Baldocchi, 2008; Oliphant et al., 2011; Williams et al., 2014) also have a long-term historical influence on the terrestrial carbon budget. Mercado et al. (2009) used JU-LES to investigate the impacts of Fdf changes on land C sink and reported an increase in C sink of 0.37 PgC  $yr^{-1}$  during 1960–1999. Compared with their estimation, our simulations show very similar impact from aerosol-caused light quality changes (0.373 PgC  $yr^{-1}$ ).

#### 4.2. The Impacts of Radiation Quality Versus Radiation Quantity

One of the key questions on aerosol impacts is whether the effect of increased diffuse radiation offsets the effect of decreasing total radiation (dimming). By investigating C fluxes at different AOD levels, Niyogi et al. (2004) found that higher AOD increases C sink at forest and cropland sites but decreases C sink at grassland sites. This could be explained by the different canopy architecture of different ecosystems. The physical mechanism underlying the diffuse radiation fertilization effect relates to shadows. With diffuse light, leaves that are packed together in a vertical sense can still receive sufficient light to maintain a high photosynthesis rate. Alternatively, under sunny conditions, leaves cast shadows that go a long way below each leaf. In short vegetation with small LAI (e.g., grassland), the shadows are more casted on soil than leaves, leading to a weaker diffuse radiation fertilization effect. In contrast, in tall vegetation with larger LAI (e.g., forests), the shadows under sunny conditions are more casted on leaves in the lower part of the

canopy, causing a deficit in radiation that cannot maintain a high photosynthesis rate on these leaves. In an extreme case, for a canopy with only one layer of leaves, all leaves should be sunlit and the increase in diffuse radiation would give no benefits to GPP. In this study, we found that substantial C sink increases are only found in regions with mean yearly LAI over 2 (Figure S5).

Apart from LAI, the overall radiation impact (quality and quantity) is suggested to also depend on whether the diffuse radiation is from the scattering of aerosols or clouds, as the latter media is thought to be less conservative regarding total downward shortwave radiation (i.e., have stronger dimming effect, Alton et al., 2008). In this study, we used a sub-grid scheme to separate the impact from cloud and aerosols. And our results shows that the GPP increase due to aerosols mainly happens in days with fewer clouds (Figure 9), which is consistent with another model-based study (Yue & Unger, 2017)

Using the NoAASW and the NoAARqlt simulations we were able to separate the effect of changes in light quantity and light quality. Globally the overall radiation impact is an increase of carbon storage of 14.7 PgC, this large enhancement indicates that the aerosol-induced radiation quality change has much stronger impact than radiation quantity changes. If we omit the difference in cloud cover, which is small at global scale (Figure 2), and consider the difference between the NoAASW and the NoAARqlt simulations as the impact of radiation quantity due to aerosols, we estimate the dimming-induced cumulative land C sink decrease to be 3.0 PgC during the study period. Regionally, our results also show that the radiation quality dominates the radiation change-caused C flux changes in most regions. Some exceptions are found in a few tropical regions such as Southeast Asia, where the amounts of anthropogenic aerosols are relatively small (Figures 7 and 8) and the forests are more light limited (Nemani et al., 2003), therefore the dimming-caused GPP and C sink decline dominates.

#### 4.3. The Main Factors Causing C Flux Changes

Only few studies have attempted to quantify the aerosol impacts on the terrestrial carbon cycle at global scale, and fewer disentangled the relative contributions of different environmental factors. By analyzing model output with a statistical method, Zhang et al. (2019) found that the increase of the land C sink in response to anthropogenic aerosols is mainly due to their cooling effect, which enhances tropical GPP and causes a stronger decline in TER than in GPP in mid latitudes. In this study, using factorial simulations with a process-based land surface model evaluated against many flux tower measurements (Zhang et al., 2020), we found similar latitudinal pattern of cooling-caused C flux changes (Figures 7 and 8) although the increased global land C sink due to this effect is much smaller than in Zhang et al. (2019). This difference in magnitude is mainly due to the internal variability of the climate system before 1880. When considering the period after 1880, the temperature impacts (10.0 PgC) detected in this study is of similar magnitude as in Zhang et al. (2019) (11.6–41.8 PgC) (Figure S4).

In contrast to temperature, Zhang et al. (2019) suggested that the aerosol-induced dimming has limited direct impacts on land C fluxes as most land ecosystems are temperature or precipitation limited rather than radiation limited (Nemani et al., 2003; Piao et al., 2009). However, this study detected a larger contribution of aerosol-induced radiation changes than temperature to global NBP changes (Figure 5b). This does not contradict the study by Zhang et al. (2019) because the large contribution is attributed to changes in light quality but not quantity (Figures 7 and 8). The light quantity changes induced by aerosols, which are generally equivalent to the radiation changes investigated by Zhang et al. (2019) in a LSM with one-stream canopy light transmission, are found to slightly decrease the GPP and NBP in intensely polluted regions (Figures 7 and 8). However, this dimming-induced decrease is not comparable in magnitude neither to the impacts of light quality changes, nor to those of temperature changes.

Compared with temperature and radiation, the impacts from changing precipitation are generally small in this study. However, this small impact from precipitation changes should be treated with caution because precipitation in current ESMs often has large uncertainties, especially in the tropics (Mehran et al., 2014). As shown by Zhang et al. (2019), if anthropogenic aerosols have a strong drying effect in very productive tropical regions, the regional C budget can be strongly affected. In IPSL-CM6A-LR climate, aerosol-induced drying is mainly found in the Sahel region and in South Asia (Figure 3b). In response, the GPP in these



regions is slightly smaller in the hist than in the NoAAprecip simulation (Figure 7c). This regional change has limited impact on the global C fluxes.

#### 4.4. The Impacts of Volcanoes

Volcanoes are a very important sporadic source of stratospheric aerosols that affect light globally and they have been reported to significantly enhance GPP at the site level due to their impact on Fdf (Gu et al., 2002). However, at the global scale, there remain no available reliable observational datasets to investigate the impacts of volcanic aerosols. The eruption of Mount Pinatubo is covered by several atmospheric inversion C flux data sets (e.g., Rödenbeck et al., 2003), but it is difficult to separate the impacts of aerosols from those of climate variations, especially given the fact that an El Nino event occurred right after the eruption. Based on models, the eruption of Mount Pinatubo is estimated to have induced an increase in land C sink of 1-2.5 PgC yr<sup>-1</sup> in the 1–2 years following the eruption, which is suggested to be strongly contributed by the Fdf changes caused by the volcanic aerosols (Mercado et al., 2009; Roderick et al., 2001). In this study, our simulations show a sudden increase of global land NBP of 3.2 PgC yr<sup>-1</sup> in response to the aerosols emitted by Mount Pinatubo and about 1.2–1.3 PgC yr<sup>-1</sup> in response to the eruptions of Santa María and El Chichón. These volcanic aerosol impacts partly explain the observed drop of atmospheric CO<sub>2</sub> growth rate after the eruption and are generally in line with the previous estimates (Mercado et al., 2009; Roderick et al., 2001). It should be noted that we only considered the change in diffuse radiation fraction in response to volcanic aerosols, but did not include the cooling and dimming caused by these eruptions. To attribute the full impacts of volcanoes, additional simulations with IPSL-CM6A-LR would be needed to provide the climate fields.

Because volcanic aerosols have a strong cooling effect on the global land surface through reducing the solar radiation, stratospheric aerosol injection technique, which mimics the eruption of volcanoes, has been proposed as a possible solar radiation management technique to mitigate global warming (Boucher et al., 2013). Some strategies consist in adjusting aerosol injections to keep the stratospheric aerosols at a high level (i.e., Tjiputra et al., 2016), in order to keep a low surface temperature. In this study, we found that the impact of such a sudden change in stratospheric aerosols has persistent impacts on land C budget after 3–5 decades (Figures 5 and 6), implying that the stratospheric aerosol injection technique may have long-lasting atmospheric carbon dioxide removal consequences, but with regionally diverse impacts. This effect needs to be considered when evaluating current solar radiation management strategies.

#### 4.5. Limitation, Implication and Future Work

In this study, we set up factorial simulations to investigate the aerosol impact during the historical period using IPSL-CM6A-LR. Although we tried to reduce the uncertainty in the ESM climate through the bias correction, there remain unavoidable uncertainties.

First, the cloud fraction from the current ESMs is still not very accurate, compared with satellite observation. The error in cloud fraction is not likely to alter the impact of diffuse radiation in the clear and cloudy parts of the gridbox alone because the radiation fluxes in the clear part of the gridbox is extracted from the look up table (Appendix A) and the diffuse radiation fertilization effect under cloudy conditions is small (Figure 9). Nevertheless, the error in cloud fraction may still affect the total impact of aerosols because it is used as a weighting factor to combine the clear and cloudy C fluxes at the subgrid scale. A recent study evaluated the cloud fraction from CMIP6 models and found that current models can generally capture the variation of cloud cover but with a SE of 0.5 compared with recent satellite observations (Vignesh et al., 2020), which is not that huge. In this study, we did not correct the bias in cloud fraction like the other climate variables because there remain no reliable cloud cover reconstructions at the preindustrial level. Even though it is possible to do the correction using present day observations, we still cannot ensure to have correct cloud cover for the early decades of our simulations and under the histNoAA scenario due to the lack of understanding on the evolution of cloud cover. To reduce this uncertainty, ESMs which can better simulate cloud cover are needed.

Second, the look up table for sub-grid radiation partitioning uses a different atmospheric radiation transfer model than the one used in IPSL-CM6A-LR, which may result in a mismatch between the total and diffuse radiation fluxes at the subgrid level. Besides this, the IPSL-CM6A-LR ESM uses a three-dimensional cloud

distribution and complex overlap assumptions to simulate radiation, the use of two-dimensional cloud cover to partition radiation to clear and cloudy subgrid in this study (Appendix A) will also cause mismatches in the subgrid level radiations. Furthermore, due to the lack of aerosol type information in IPSL-CM6A-LR to match the aerosol types in the look-up table, we have to assume a single type of aerosol in each region, which introduces some uncertainty in the Fdf.

Third, as shown in our results (Figures 5 and S1), the internal variability of the climate system can also affect the C fluxes. To eliminate this impact, ensemble simulations with IPSL-CM6A-LR starting from slightly different initial state are needed to provide the histNoAA climate with different internal climate variabilities. With limited resources, we are not able to do such simulations now. Nevertheless, the impacts from climate internal variability are not likely to change our main conclusions because the magnitude of the cumulative NBP change due to climate internal variability (about 6 PgC) is much smaller than the aerosol impact we detected after the 1950s (over 20 PgC). In addition, the GPP and cumulative NBP from the IPSL-CM6A-LR historical ensemble simulations (with different climate internal variability) shows high consistency, and have standard deviations less than 2.5 PgC in most of the years (Figure S6), which is much less than the aerosol impacts detected in this study.

In spite of the bias correction of climate variables, there remain uncertainties in the aerosol-induced climate changes (i.e., difference between historical and histNoAA). As shown by Zhang et al. (2019), the ESM uncertainty in aerosol-induced climate changes, especially precipitation changes, can potentially cause large uncertainty in the C fluxes. However, the histNoAA is not a mandatory experiment in CMIP6, and few models have performed it. Also, not all CMIP6 data have been published currently. It thus remains difficult to quantify the uncertainty in this effect from a multimodel perspective at this moment. One may argue that many CMIP6 ESMs have the hist-aer experiments (i.e., only consider the changes of tropospheric aerosols but keep the other anthropogenic forcings constant) and pairing these experiments with their corresponding CMIP6 piControl experiments (no anthropogenic forcings considered) may also be used to quantify the aerosol impacts. However, the piControl experiments have different climate baseline compared with the historical experiments, thus the aerosol impacts estimated from hist-aer and piControl climate must be biased from the real aerosol impacts.

Besides the forcing field, uncertainties may also arise from the land surface model used. In this study we used only ORCHIDEE\_DF to investigate the Fdf and climate effects. Although the capacity of ORCHIDEE\_DF to capture the diffuse light fertilization effect has been carefully evaluated at a large number of flux towers (Zhang et al., 2020), there remain structural uncertainties due to the other processes in ORCHIDEE\_DF. To understand this uncertainty, a multi-model study on aerosol impacts would be needed to confirm the considerable impact from aerosol-induced radiation quality changes found in this study. Candidate LSMs for such a multimodel study must have a two-stream canopy radiation transmission to distinguish direct and diffuse light, as it is the case of the JULES (Mercado et al., 2009), YIBs (Yue & Unger, 2015), iTem LSM (Chen & Zhuang, 2014), and CLM (Oleson et al., 2013) models. Besides the uncertainties in current model processes, the lack of other physiological processes may also bias the results. For instance, the ORCHIDEE\_DF model includes no nutrient limitation on vegetation growth, which means the deposition of nutrients along with the deposition of aerosols is not considered in our simulations. It has been suggested that N deposition can be a main factor causing land C sink in some ecosystems (e.g., Magnani et al., 2007; Vuichard et al., 2019). Understanding how this effect interacts with aerosol-induced climate and Fdf requires land surface models with additional complexity.

It should be also noted that the aerosol impacts investigated in this study lack a range of potentially important feedbacks. The increased land C sink will slow the growth of the atmospheric  $CO_2$  concentration, which can decrease the  $CO_2$  fertilization effect on vegetation and decrease land C sink. Meanwhile, the slowed growth rate of atmospheric  $CO_2$  will slow the warming and increase land C sink. Besides these mechanisms, the changes of land C cycle may also alter the land surface albedo and roughness by affecting the vegetation, which will cause biophysical feedbacks to climate. To investigate these feedbacks, coupled simulations with ESMs that consider the diffuse light impacts are needed. Because the climate bias correction technique cannot be used in coupled simulations, the ESMs need to be carefully calibrated.



## 5. Conclusions

In summary, we performed and reported here a set of factorial simulations using the ORCHIDEE LSM forced with climate fields from the IPSL-CM6A-LR ESM under different configurations. We calculated the differential impact of diffuse and direct radiation in the cloudy and clear gridbox fractions, in order to understand the impact of anthropogenic aerosols on terrestrial C fluxes during the period since 1850. We found that, by the year 2014, anthropogenic aerosols cumulatively enhanced land C sink by 22.6 PgC, out of which 78% is attributed to the impact of aerosol-induced radiation quality changes. This enhancement of radiation quality changes is much stronger than the dimming caused by aerosols. Although the cooling due to anthropogenic aerosols on average only slightly enhances the global land C sink, it has considerable impacts on its temporal evolution and at different latitudes. Compared with radiation and temperature, precipitation changes due to aerosols have smaller impacts on C fluxes in our study.

The diffuse radiation impact on land C sink detected in this study is of considerable magnitude. This large impact highlights the necessity of considering the radiation quality in simulating land C fluxes, especially for the projection of future climate under different aerosol emission scenarios, as we should remind that the climate-carbon feedback is one of the most important ones in the Earth system (Friedlingstein et al., 2006, 2014). Unfortunately, the direct and diffuse radiative fluxes are only distinguished and adequately treated in a limited number of LSMs. There are almost no studies evaluating this process in current LSMs at large scale. Such studies are thus urgently needed to better quantify current aerosol impacts and project future C fluxes under different aerosol emission scenarios.

# Appendix A: Radiation Partitioning and Diffuse Radiation Calculation at Subgrid Level

At each time step, each gridbox is first divided into a clear and a cloudy part according to the total cloud fraction (clt) of this gridbox (Figure 1). In the clear sky sub-grid area, the incoming direct (SWdown<sub>clear,dir</sub>) and diffuse (SWdown<sub>clear,dif</sub>) shortwave radiation at the surface are obtained considering tropospheric and stratospheric aerosol optical depth at 550 nm, as well as solar zenith angle from a look up table. The look up table is built with an atmospheric radiative transfer model named Streamer (Key & Schweiger, 1998) as initially described in Mercado et al. (2009). Using SWdown<sub>clear,dir</sub> and SWdown<sub>clear,dif</sub> we are able to calculate the sum of their counterparts in the cloudy sub-grid area (SWdown, which is provided in IPSL-CM6A-LR climate:

$$SWdown = \left(SWdown_{clear,dir} + SWdown_{clear,dif}\right) \times \left(1 - clt\right) + \left(SWdown_{cloudy,dir} + SWdown_{cloudy,dif}\right) \times clt, (A1)$$

where  $SWdown_{cloudy,dir}$  and  $SWdown_{cloudy,dir}$  represent the direct and diffuse radiative fluxes of the cloudy subgrid area. Here, we assume  $SWdown_{cloudy,dir}$  to be zero following Mercado et al. (2009). This is a good assumption because the optical depth of clouds is generally larger than a few units except for thin high-level clouds. The  $SW_{cloudy,dir}$  can then be estimated as:

$$SWdown_{cloudy,dif} = \frac{SWdown - (SWdown_{clear,dir} + SWdown_{clear,dif}) \times (1 - clt)}{clt},$$
 (A2)

where SWdown and clt come from the forcing files while SWdown<sub>clear,dir</sub> and SWdown<sub>clear,dif</sub> come from the look up table. It should be noted that as ORCHIDEE\_DF has a time step of 30 min but the forcing fields have coarser temporal resolution, we have to interpolate all forcing data into 30 min in ORCHIDEE\_DF. Furthermore, the IPSL-CM6A-LR model, which has generated the forcing fields, and the look up table use different radiative transfer scheme and the IPSL-CM6A-LR radiation scheme relies on the full 3D structure of clouds. This causes unavoidable mismatch between the different terms of Equation A2, which may occasionally lead to unreasonable negative SWdown<sub>cloudy,dif</sub> or larger SWdown<sub>cloudy,dif</sub> than SWdown-clear,dir + SWdown<sub>clear,dir</sub>. In the first case, the SWdown<sub>cloudy,dif</sub> is set as zero and SWdown<sub>cloudy,dif</sub> is capped



to  $SWdown_{clear,dir} + SWdown_{clear,dif}$  and then  $SWdown_{clear,dir}$  and  $SWdown_{clear,dif}$  are linearly scaled to equilibrate Equation A1.

## Appendix B: Bias Correction of Climate Data

As climate baseline may affect the C fluxes in our simulation, we used the observation-based CRUJRA climate (Harris, 2019; Harris et al., 2014; Kobayashi et al., 2015) as the reference to remove the bias in the original IPSL-CM6A-LR climate, following the method described in Zhang et al. (2019):

For Tmax, Tmin, Wind\_N, Wind\_E, and Ps, the bias-corrected data is calculated as:

$$V_{\rm corr}\left(t,m\right) = V_{\rm mod}\left(t,m\right) - \Delta V\left(m\right),\tag{B1}$$

$$\Delta V(m) = V_{\text{base,hist}}(m) - V_{\text{base,obs}}(m), \tag{B2}$$

while for Precip, SWdown, LWdown, and Qair, the bias-corrected data is calculated as:

$$V_{\rm corr}\left(t,m\right) = \frac{V_{\rm mod}\left(t,m\right)}{\Delta V(m)},\tag{B3}$$

$$\Delta V(m) = \frac{V_{\text{base,hist}}(m)}{V_{\text{base,obs}}(m)},\tag{B4}$$

where  $V_{\text{corr}}(t,m)$  is the bias-corrected climate at day *t* of month *m*,  $V_{\text{mod}}(t,m)$  is the original IPSL-CM6A-LR climate at time *t* of month *m*,  $V_{\text{base,obs}}(m)$  is the climate baseline from CRUJRA in month *m*, calculated as the mean climate in month *m* during the first 10 years of this data set (1901–1920).  $V_{\text{base,hist}}(m)$  is the modeled climate baseline of month *m*, calculated during 1850–1869 using the average of the *historical* climate.

As shown by Zhang et al. (2019), due to the weak perturbation of anthropogenic activities before 1900, the climate did not change much during this period. Therefore, the mismatch in baseline period will not remarkably alter our results.

#### Data Availability Statement

The authors thank Guillaume Levavasseur for his help with the curation of the IPSL-CM6A-LR data. The IPSL-CM6A-LR data used in this study can be accessed through https://esgf-node.llnl.gov/search/cmip6/ and the CRUJRA data can be obtained in the reference Harris (2019).

#### References

Alton, P. B. (2008). Reduced carbon sequestration in terrestrial ecosystems under overcast skies compared to clear skies. Agricultural and Forest Meteorology, 148(10), 1641–1653. https://doi.org/10.1016/j.agrformet.2008.05.014

- Alton, P. B., North, P. R., & Los, S. O. (2007). The impact of diffuse sunlight on canopy light-use efficiency, gross photosynthetic product and net ecosystem exchange in three forest biomes. *Global Change Biology*, 13(4), 776–787. https://doi.org/10.1111/j.1365-2486.2007.01316.x Boucher, O., Randall, D., Artaxo, P., Bretherton, C., Feingold, G., Forster, P., et al. (2013). Clouds and aerosols. In *Climate change 2013: The*
- physical science basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change (pp. 571–657). Cambridge: Cambridge University Press.
- Boucher, O., Servonnat, J., Albright, A. L., Aumont, O., Balkanski, Y., Bastrikov, V., et al. (2020). Presentation and evaluation of the IPSL-CM6A-LR climate model. *Journal of Advances in Modeling Earth Systems*, *12*(7), e2019MS002010. https://doi.org/10.1029/2019ms002010 Chen, M., & Zhuang, Q. (2014). Evaluating aerosol direct radiative effects on global terrestrial ecosystem carbon dynamics from 2003 to 2010. *Tellus B: Chemical and Physical Meteorology*, *66*(1), 21808. https://doi.org/10.3402/tellusb.v66.21808
- Cheng, S. J., Bohrer, G., Steiner, A. L., Hollinger, D. Y., Suyker, A., Phillips, R. P., & Nadelhoffer, K. J. (2015). Variations in the influence of diffuse light on gross primary productivity in temperate ecosystems. Agricultural and Forest Feteorology, 201, 98–110.

#### Acknowledgments

The authors acknowledge support from European Research Council Synergy project SyG-2013-610028 IMBAL-ANCE-P, H2020-EU.3.5.1. 4C project (Grant No. 821003), European Commission's Horizon 2020 CRESCENDO project (Grant No. 641816) and the French ANR project China-Trend-Stream.



- Choudhury, B. (2001). Estimating gross photosynthesis using satellite and ancillary data approach and preliminary results. *Remote Sensing of Environment*, 75(1), 1–21. https://doi.org/10.1016/s0034-4257(00)00151-6
- Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the coupled model intercomparison project phase 6 (CMIP6) experimental design and organization. *Geoscientific Model Development*, 9, 1937–1958. https://doi. org/10.5194/gmd-9-1937-2016
- Friedlingstein, P., Cox, P., Betts, R., Bopp, L., von Bloh, W., Brovkin, V., et al. (2006). Climate-carbon cycle feedback analysis: Results from the C4MIP Model Intercomparison. *Journal of Climate*, 19(14), 3337–3353. https://doi.org/10.1175/jcli3800.1
- Friedlingstein, P., Meinshausen, M., Arora, V. K., Jones, C. D., Anav, A., Liddicoat, S. K., & Knutti, R. (2014). Uncertainties in CMIP5 climate projections due to carbon cycle feedbacks. *Journal of Climate*, 27(2), 511–526. https://doi.org/10.1175/jcli-d-12-00579.1
- Gu, L., Baldocchi, D., Verma, S. B., Black, T., Vesala, T., Falge, E. M., & Dowty, P. R. (2002). Advantages of diffuse radiation for terrestrial ecosystem productivity. *Journal of Geophysical Research*, 107(D6). ACL1–ACL23. https://doi.org/10.1029/2001jd001242
- Harris, I. (2019). CRU JRA v1.1: A forcings dataset of gridded land surface blend of Climatic Research Unit (CRU) and Japanese reanalysis (JRA) data. Centre for Environmental Data Analysis. https://doi.org/10.5285/13f3635174794bb98cf8ac4b0ee8f4ed
- Harris, I., Jones, P. D., Osborn, T. J., & Lister, D. H. (2014). Updated high-resolution grids of monthly climatic observations—The CRU TS3.10 Dataset. *International Journal of Climatology*, 34(3), 623–642. https://doi.org/10.1002/joc.3711
- Healey, K. D., Hammer, G. L., Rickert, K. G., & Bange, M. P. (1998). Radiation use efficiency increases when the diffuse component of incident radiation is enhanced under shade. Australian Journal of Agricultural Research, 49, 665–672. https://doi.org/10.1071/a97100
- Hollinger, D. Y., Kelliher, F. M., Byers, J. N., Hunt, J. E., McSeveny, T. M., & Weir, P. L. (1994). Carbon dioxide exchange between an undisturbed old-growth temperate forest and the atmosphere. *Ecology*, 75(1), 134–150. https://doi.org/10.2307/1939390
- Jones, C. D., Cox, P. M., Essery, R. L., Roberts, D. L., & Woodage, M. J. (2003). Strong carbon cycle feedbacks in a climate model with interactive CO2 and sulphate aerosols. *Geophysical Research Letters*, 30(9), 1479. 2003. https://doi.org/10.1029/2003GL016867
- Kanniah, K. D., Beringer, J., Tapper, N. J., & Long, C. N. (2010). Aerosols and their influence on radiation partitioning and savanna productivity in northern Australia. *Theoretical and Applied Climatology*, 100(3–4), 423–438. https://doi.org/10.1007/s00704-009-0192-z
- Key, J. R., & Schweiger, A. J. (1998). Tools for atmospheric radiative transfer: Streamer and FluxNet. Computers & Geosciences, 24(5), 443–451. https://doi.org/10.1016/s0098-3004(97)00130-1
- Knohl, A., & Baldocchi, D. D. (2008). Effects of diffuse radiation on canopy gas exchange processes in a forest ecosystem. Journal of Geophysical Research. 113. G02023. https://doi.org/10.1029/2007JG000663
- Kobayashi, S., Ota, Y., Harada, Y., Ebita, A., Moriya, M., Onoda, H., et al. (2015). The JRA-55 reanalysis: General specifications and basic characteristics. *Journal of the Meteorological Society of Japan*, 93(1), 5–48. https://doi.org/10.2151/jmsj.2015-001
- Krinner, G., Viovy, N., de Noblet-Ducoudré, N., Ogée, J., Polcher, J., Friedlingstein, P., et al. (2005). A dynamic global vegetation model for studies of the coupled atmosphere-biosphere system. *Global Biogeochemical Cycles*, 19(1), GB1015. https://doi.org/10.1029/2003GB002199
- Lardy, R., Bellocchi, G., & Soussana, J.-F. (2011). A new method to determine soil organic carbon equilibrium. *Environmental Modelling & Software*, 26(12), 1759–1763.
- Magnani, F., Mencuccini, M., Borghetti, M., Berbigier, P., Berninger, F., Delzon, S., et al. (2007). The human footprint in the carbon cycle of temperate and boreal forests. *Nature*, 447(7146), 849–851. https://doi.org/10.1038/nature05847
- Mahowald, N., Lindsay, K., Rothenberg, D., Doney, S. C., Moore, J. K., Thornton, P., et al. (2011). Desert dust and anthropogenic aerosol interactions in the community climate system model coupled-carbon-climate model. *Biogeosciences*, 8(2), 387–414. https://doi. org/10.5194/bg-8-387-2011
- Mehran, A., AghaKouchak, A., & Phillips, T. J. (2014). Evaluation of CMIP5 continental precipitation simulations relative to satellite-based gauge-adjusted observations. Journal of Geophysical Research: Atmospheres, 119(4), 1695–1707. https://doi.org/10.1002/2013jd021152
- Mercado, L. M., Bellouin, N., Sitch, S., Boucher, O., Huntingford, C., Wild, M., & Cox, P. M. (2009). Impact of changes in diffuse radiation on the global land carbon sink. *Nature*, 458(7241), 1014–1017. https://doi.org/10.1038/nature07949
- Misson, L., Lunden, M., McKay, M., & Goldstein, A. H. (2005). Atmospheric aerosol light scattering and surface wetness influence the diurnal pattern of net ecosystem exchange in a semi-arid ponderosa pine plantation. Agricultural and Forest Meteorology. 129, 69–83.
- Nemani, R. R., Keeling, C. D., Hashimoto, H., Jolly, W. M., Piper, S. C., Tucker, C. J., et al. (2003). Climate-driven increases in global terrestrial net primary production from 1982 to 1999. *Science*, 300(5625), 1560–1563. https://doi.org/10.1126/science.1082750
- Niyogi, D., Chang, H. I., Saxena, V., Holt, T., Alapaty, K., Booker, F., et al. (2004). Direct observations of the effects of aerosol loading on net ecosystem CO2 exchanges over different landscapes. *Geophysical Research Letters*, 31, L20506. https://doi.org/10.1029/2004GL020915
- Oleson, K., Lawrence, D., Bonan, G., Drewniak, B., Huang, M., Koven, C., et al. (2013).*Technical description of version 4.5 of the Community* Land Model (CLM), NCAR Technical Note: NCAR/TN-503+ STR. Boulder, CO: National Center for Atmospheric Research (NCAR).
- Oliphant, A., Dragoni, D., Deng, B., Grimmond, C., Schmid, H.-P., & Scott, S. (2011). The role of sky conditions on gross primary production in a mixed deciduous forest. Agricultural and Forest Meteorology. 151(7), 781–791.
- Piao, S., Ciais, P., Friedlingstein, P., de Noblet-Ducoudré, N., Cadule, P., Viovy, N., & Wang, T. (2009). Spatiotemporal patterns of terrestrial carbon cycle during the 20th century. *Global Biogeochemical Cycles*, 23. https://doi.org/10.1029/2008GB003339
- Piao, S., Sitch, S., Ciais, P., Friedlingstein, P., Peylin, P., Wang, X., et al. (2013). Evaluation of terrestrial carbon cycle models for their response to climate variability and to CO2 trends. *Global Change Biology*, 19, 2117–2132. https://doi.org/10.1111/gcb.12187
- Rödenbeck, C., Houweling, S., Gloor, M., & Heimann, M. (2003). CO2 flux history 1982–2001 inferred from atmospheric data using a global inversion of atmospheric transport. Atmospheric Chemistry and Physics, 3, 1919–1964. https://doi.org/10.5194/acp-3-1919-2003
- Roderick, M. L., Farquhar, G. D., Berry, S. L., & Noble, I. R. (2001). On the direct effect of clouds and atmospheric particles on the productivity and structure of vegetation. *Oecologia*, 129(1), 21–30. https://doi.org/10.1007/s004420100760
- Tjiputra, J. F., Grini, A., & Lee, H. (2016). Impact of idealized future stratospheric aerosol injection on the large-scale ocean and land carbon cycles. Journal of Geophysical Research: Biogeosciences, 121(1), 2–27. https://doi.org/10.1002/2015jg003045
- Vignesh, P. P., Jiang, J. H., Kishore, P., Su, H., Smay, T., Brighton, N., & Velicogna, I. (2020). Assessment of CMIP6 cloud fraction and comparison with satellite observations. *Earth and Space Science*, 7, e2019EA000975. https://doi.org/10.1029/2019EA000975
  - Vuichard, N., Messina, P., Luyssaert, S., Guenet, B., Zaehle, S., Ghattas, J., et al. (2019). Accounting for carbon and nitrogen interactions in the global terrestrial ecosystem model ORCHIDEE (trunk version, rev 4999): Multi-scale evaluation of gross primary production. *Geoscientific Model Development*, 12, 4751–4779. https://doi.org/10.5194/gmd-12-4751-2019
  - Williams, M., Rastetter, E. B., Van der Pol, L., & Shaver, G. R. (2014). Arctic canopy photosynthetic efficiency enhanced under diffuse light, linked to a reduction in the fraction of the canopy in deep shade. *New Phytologist. 202*(4), 1267–1276.
  - Xie, X., Wang, T., Yue, X., Li, S., Zhuang, B., & Wang, M. (2020). Effects of atmospheric aerosols on terrestrial carbon fluxes and CO<sub>2</sub> concentrations in China. Atmospheric Research, 237, 104859. https://doi.org/10.1016/j.atmosres.2020.104859



- Yue, X., & Unger, N. (2015). The Yale interactive terrestrial Biosphere model version 1.0: Description, evaluation and implementation into NASA GISS ModelE2. *Geoscientific Model Development*, 8(8), 2399–2417. https://doi.org/10.5194/gmd-8-2399-2015
- Yue, X., & Unger, N. (2017). Aerosol optical depth thresholds as a tool to assess diffuse radiation fertilization of the land carbon uptake in China. *Atmospheric Chemistry and Physics*, *17*, 1329–1342. https://doi.org/10.5194/acp-17-1329-2017
- Zhang, Y., Bastos, A., Maignan, F., Goll, D., Boucher, O., Li, L., et al. (2020). Modeling the impacts of diffuse light fraction on photosynthesis in ORCHIDEE (v5453) land surface model. *Geoscientific Model Development*, *13*, 5401–5423. https://doi.org/10.5194/gmd-13-5401-2020
- Zhang, Y., Boucher, O., Ciais, P., Li, L., & Bellouin, N. (2021). How to reconstruct aerosol-induced diffuse radiation scenario for simulating GPP in land surface models? An evaluation of reconstruction methods with ORCHIDEE\_DFv1.0\_DFforc. Geoscientific Model Development, 14, 2029–2039. https://doi.org/10.5194/gmd-14-2029-2021
- Zhang, Y., Goll, D., Bastos, A., Balkanski, Y., Boucher, O., Cescatti, A., et al. (2019). Increased global land carbon sink due to aerosol-induced cooling. *Global Biogeochemical Cycles*, 33(3), 439–457. https://doi.org/10.1029/2018gb006051